Author's response to reviews

Title: Mycotic Aneurysm of the Posterior Tibial Artery - A rare complication of bacterial endocarditis: a case report

Authors:

  Sandeep Patel (sandeep99@doctors.net.uk)
  Nigel D'Souza (nige@doctrs.net.uk)
  Shashank Gurjar (syguriar@gmail.com)
  Jim Hewes (jimhewes@hotmail.com)

Version: 5 Date: 7 August 2008

Author's response to reviews:

Requested changes made to document. One image removed.